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AUTHDA: GeJoS, Rerry (JoKe Blacow)
PROSLENM: To continue presantetion of s long renge plan to improve the
production ofAroclor, Pyroclor and Diphenyl at Newport.
SUNRARY
o) Areclors

To mest the Five Yesr Salas Forscest, Aroclor cepsoity needs to
be sxpended sbove the present 9.3§ W 1be per annum stendard. The problem is to
sohisve this increase consiestent with improving the quslity of the finished
product to schieve the Company goale. Thass apply to the elsctrical gredes of
1242 and 1254 and have besn eset up for Power Fector, Resietivity, Wstsr Content,
Stability end Colour.

Potential chlorinstion cepscity is sdequate for the t
period but the incressed gessing retes Tequired will necessitate installetion of
Brink Miet Eliminstors to cut down entrainment end so svoid blockeges end meintein
yielde The jecket cooling ie sleo insdequste and external fecilitiss will bs
required. With these sdditiona, operetion st the seme levels se M.C. will bs
poseible.

A ospsoity of 13 [] 1bs/annum has besn demonstrated in etill triele
with betch opsrstion but distillste colour is bersly sccepteble. It dows not
west the quality goal. The provision of improved entreinment ssperstion end
heet input fscilitiss ere being investigsted, Although it le poseible that the
existing still may be sdeguete for production beyond 1970 with extensive
modificetions, the installetion of & sscond system will be required to snsure
this. The resulting capacity would be sdequate to cover 1972 production eith
the 25% excess sveilability recommended at the St. Louis Quality Mesting in
Februery, 1966.
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Evaluation of the propoesd alternative trestment meteriel (bauxite) for
improved elsctricel properties will be umttnund » starting with e plent triel
under beteoh conditions.
b) Byrogler
Production Facilitims ers edequsts for 1967/8 requirements.
Minor epecificetion problems will be investigeted.
o) Dighenyl

Sub ject to epproval of the C.A.R. the inetalletion of the tubuler
unit will be procseded with, to meet diphenyl requiremente end evoid the
neceseity to import M,C. Diphenyl.
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BECAPTTULATION
Savioge 8.1,
A0 Statye B i @ Sopital Hotse
rink Miet CoAsR, Submiesion - £6000 £5000 £3000  Sevinge from
Eliminetors let. querter '67 ) yield loes elim-
: ination.
, ) Estd. 2% at high
' chlorination rates

Sentowex On 1ine Feb. '67 £4000° £4000 £4000 25000 Sevings from
Fusl Uss ' roduced oil usage

) :. : & weste disposel
Diphanyl Colols in - - 295000 E30S000 Sevinge from
Tubuler prepsretion s ' reduced benzens
Unit ‘ uenge & flske

importe.
Productivity
Aroclor 45000 £10000 230000 -
Pyzoclor AAGGD L7000 AODGD -
R M3000 27000 A143000
CURRENT STATUS & MAJOR PRODLENS
1. DProbles Argas
a) Aroclore

b)

s)

The mejor process problem sres remsins, ths production of low powsr
fector slectrical grede Aroclor, in particuler Aroclor 1242. Associsted
with this problem s the difficulty still encountared in obtaining
relisble wegsursments of thie psrameter.

The prooess bottlsnsck occurs in the distilletion step. Cepscity

is ot pressnt limitsd by the ability of the etill to produce e distillete
of scoepteble colour.

Pyreclor

No mejor problems ars involved in Pyroclor production. Some
minor difficultise are etill sesocieted with retreatment of vetches.

Riphenyl

There are no msjor problems eseociated with Diphenyl.
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A)

Srocese Yield
Areclogs

Principle rew meterials, chlorine snd diphenyl, sre meds on eite
spsrt from fleke import required to supplement diphenyl production (et &% of
total requirement {n 1966)

No messsurement fecilitiss exist st presant to pesrmit sn sccurets
estimets of yielde

MeCole usagee closs to those ofM.Cs have been shown previously.

Current usegee aret-

Theoretical Usege Stenderd 04fference Valus et
1b/1000 1b F.P. _ 1967 Requirements
1242 BA9 904 24,300
1248 970 1038 450
1254 1090 1165 2,300
1260 1210 1300 1,100 ]
1262 1250 1338 60
I _Basis Chloine 3:0d/ab = = |
1242 892 620 5,700
1248 838 560 500
1254 aT8 498 1,900
1260 45 437 680
1262 398 lnv 30
—Dasie Dipheny) o3l

MONS 056826



-5.

Yield losses due to entreinment from the chlorinstors will be
min{mised by inetslletion of Brink Mist Eliminetors as part of the expsnsion
progremme. The capitel cost is spproximetely £3,000.

b) Syreclor

The bulk of pyroclor production is Pyrocior P.P.0. (Nominelly
60% 1260, 40K T.C.8, with Phenoxy Propene Oxide ges scavenger et the 0.2%
level). In addition, e small demand exists for Special Pyroclor(75%
1254, 25% TC8) Aroclor 1260 (& 1254) is trensferred on a standerd besis
of 620 1b/1000 1b F.P. (760 1b/1000 1b in the cass of Special Pyroclor)

4964 1963 = 1966  Standard

TeCoBe usesges 413 419 402 415
1b/1000 1be F.P.

The improvement hae resultsd from bulk handling of T.C.8. end
relexetion of an unneceesserily tight resistivity speciflcetion.

¢) Righenvl

Diphenyl requiremente are in excess of the cepacity of the Leed
Pot unite. Operation to maximise benzene convereion per pass and hence
production hee resulted in highsr benzene usages.

A8 166 Staodaxd

Benzene usege . 157 160,.2 162
gel1/1000 1b F.P,

If inetelletion of Tubuler Diphenyl 011 ie epproved, Benzene savinge
of 217,000 (et the 5.5 A 1b lavel) will scorue from lower conversion level
operetion.

The existing. hydrogen compressor le undereized by sppreximetely
20% end e lerger unit will be reguired to achisve maximum benzene recovery
if the Leed pots installetion is meinteined. The ospital cost for e new
compressor will be spproximetely £2,500.

3o
-)mmwmxmmxxm

Sales requirsments * 10.2 11.S 12,8 14,2 15,8 17.6
M 1b,

lmlr-m. te with 12.7 14.4 15.6 17.7 19.7 22,0
oxtres 29% surge

evailebility W 1b
* Requiremsnts include tranefer to Pyroclor
Pressnt Standerd Capeeitys 9.35 A lbs

Pressnt relizeble ospacitys 10.0 M 1lbe (assumsa use of semi-continuous
fecilities)
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Capacity incresses sre requirsd fn the srses of chlorinastion end
distilletion to mest the forscast figures. The 25X excess cepacity hee been
cecommended for ell Arvclor plante (8t. Loule Quality Meeting Februery, 1966)
to allow for peak demsnde.

1) Shioxdnation
Existing cepecity 12 W 1b
Sotential cepecity 20 W1b
Cepttal 28000

Inetalletion of e 4" chlorine transfer line in October 1968
allows simultensaue gessing on all thres chlorinstore st levels currently in
operetion by MsCe (up to 1500 1b/hr). At the higher chlorinetion retas the
p t cyclone tore will be unsble to deal efficiently with the
increassd entreinment leed. Design speratien hee in any ceee not been echieved
due to pluggepe problem with ferric chleride. It ie proposed to install 8rink
Mist sliminetors to minimiss sntrainment 10sses. Preliminery trisls et high
chiorinetion rates have shown the oooling system (water jeckets) to be
inadequets. CExternal oveling will be providedi This will teke the form of
air osolers aimiler to W.C. installstiona! With thie spplicetion of N.C.
technology, chlorinetd pereti will have sufficient capscity for the next

S years.
11) Riatillation
Stenderd Cepsoity 9:35 B 1b/snnum
). Potentiel Capscity Intermediste 14 W 1b
(existing system) Potential 1780

Plant trials have shewn that betch cycls tises, giving cepecities
of up to sbout 13 M 1b per ennum of Argoler 1242 are evaileble with the exieting
oquipment. At levele in excess of 10 R 1b, loss of entreinment seperatlon efficiency
leads %0 o0 & in preduct osl A brosd correlation betwesn betch oolour
end distilletion rate exists but with oensidersble varietion betwsen betchee.

Typioal results for Arcoler 1242 are s«

Coil ATOC 9 10 13

Cycle time hes, 0.3 7.8 6.2

Capeoity (87% gil.'r.) 94 10,0 12.8
be

Colour hazen 30/40 40/50 40/%0

In norwal operetien s maximum AT of 99C ie used on Aroclor 1242
For the higher Avoclers (1260 end 1262) & higher spreed 1e used (up to 149C) to
sllow for the greater hest lessee associated eith the higher & t of operati

Tha quality goals set up in February, 1966 require e finished produot
colour of 10 APHA, To schisve this, e dietilletion betch colour of lese than 20 APKA
is necessary. Thus a mers efficient entreinment ssperstor will be resquired if the
still is to give capacities over 10 M 1b while etill mesting F.P. colour requirsments.
Estimated cepital coet Ls epproximetsly £2,000.
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Instellation of increased F.P, storege for Aroclor 1242f {e scheduled
on=line in April 1967. The provision of thie fecility will sllow incresesd use to
be made of semi-continusus distillation. With improved entrainment ssparatstion to
sllow reslizetion of the capscity of ths existing hesting system, & capecity of
approx. 14 M 1b, will be sveileble. (Assumes 40-S0% Aroclor 1242 production mads
on semi-.continuous nponttcm)

Further cepscity increese will require new equipment to {ncresss
heat input. Although increessd coil ares (2" to 3") end s higher pumping rate
(cost £1,000) will be considered es the first spproach, experience st Krummrich
hee indicsted thet e eimple increess in coil dismeter will not give s significantly
increesed hest input. In thie event s new hsster will be required at a ocepitel
cost of £5000. Thie step may become necessary st en esrlier lcm if conelstent
opecation of ths pressnt set up with the higher coil t dirf
leads to excessive problems with higher retes of ooil plugq-qa & fallure. High
temperature differences will also lead to s decresss in yleld eesocietsd sith
increassed degradetion of the still contents in the coil.

Associsted squipwent (Aroclor condenser, Aroclor cooling weter
has ity up to spproxiestsly 18 ¥ 1be/enoum. Replacement of
these items will lnvol.vn » capital expenditurs of £2,000

2) Potentiel Cepecity (Duplicats system) 20 N 1b/annum
Capital £4%,000

As an altemnetive to mest the leng term capacity incresss,installation
of a sscond distilletion unit eimilar to the existing one, cen be installed snd
space haa been left in the plent layout fer this. This will giva excess caspacity
over the forscest period and the risk of nen~fulfiliment of sales requirsments eill
be eliminated.

In view of the expending salss forscests, the sccent on quality of
slectriosl greds Arvclore with respect to slectrical properties snd stadility
together with the high quality steclor produced by our competitors end the
recommendation that sach Monsente lecetien should heve 25X surge cepacity, it is
poseible thet thies should be the recesmended coures.

114) Storage

Installation of incressed F.P. (12 42 )} etorege end impraved bulk
loeding facilities is in hend.

Completion ¢ 4/67
Cepital s £21,800

Further sterege cepacity (an extre 12,000 gelle) will bs required during
1969 to meet incressed ssles volume, Cost will bs epproximetely £7,300
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AVERAGE Car °*
CHLORINATC

Charge . Ib.
Chlorination batch lb.

Charge time hra
Chlorinetion
Oischarge time

Cycls time

Capacity @ 87% 0.5.T.K 1b/
en. (per chlorinator)

Aversge Cepscity (plr

esent_(Aver. 4 8-
1142 1148 1154 1160 1162
4650 4650 4650 €650 4650

8020 8950 10,100 11620 12220

0.3 0.3 0.3 0.3 0.3
13.0 15.5 17.5 22.0 23.5
0.5 0.5 0.5 0.5 0.5
13.8 16.3 18.3 22.8 24.3

4.1 3.9 3.8 3.5 3.5

I Y

t_(Intersedist
142 1148 1154 1160

no chenos
no chehge

i162

no change
9.2 11.4 14.4
no change

18.2 19.7

10.0 12.2 15.2 19.0 20.S5

S.9 S.4 4.9 4.4 43

Terget tent’ -1

1142 1148 1160 1162
no change
no chanje
no change

7.0 8.5 10.5 13.0 14
no chanje

7.8 8.3 11.3 13.8, 14.

7.5 7.0 6.5 6.0 6.

3.9 Wib (1967 product ratio)

5.6 % 1b (1969 Product ratic)

7.2 M1b (1972 Product Ratio)

chlotinstor)
Total Chlorinstion - - - IS
Cepacity. 11.7 R lbs/snnum 16,80 Lbe/ennus 21.6 W lts/snnum o
0
L -
Q
Intermediets Cycle u-u bnod on 3 hours operstion st 600 1b/hr; ressinder st 800 1b/hr
Potengtisl * opereticn es et W.C. 2 hours @ 600 ib/hr 2
1 hour @ 800 xo/m S
Ressinder @ 1200 1b/hr x
OwMTING:-
feintenence 6%
Pracess
Gthsr 5%




Dgw Tanks 1142 1148 1154 1160 162

Cherges ae per chlorinator batoh

Charge time hre. 0.5 0.5 0.5 0.5 0.5
, Blom time hre 4.0 4,0 4,0 4,0 4.0
Discherge time hrs. 0.3 0.3 0.3 0.3 0.3
Cycle tims hrs. 4.8 4.8 4,8 4,8 4.8
Capscity 8 97% 0.8.7. W 1b 14,2 15.7 17,7 20,3 2.4
Per blow tenk.
Aversge cepecity (1967 product retis) 15.8 K 1b/annum
per Blow tenk
Total capecity 47,4 W 1b/ennum
' Down time
Maintenence 1%
Process 1% -
Other 1%
Air-blowing capsoity ie gquits sdeguate for the f reguirements and no

sdditione ers nsceesary.
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Djstillstion (1) Batch Opsration
1242
Charge (Average 875 gall) 1b 11000
Distillation 8stch 1b 10700
Time cycle (tiwes in hours)
Charge/hest up 2.0
Recycle 0.25
Distillstion 6.0
Bottoms withdrawsl * 0.15
Cycle time 6.4
Cepacity @ 87% 0.5.T, W 1b/annum 9.3

Actusl Dietillstion Capacity

M lbs/snnum

* Bottome withdrawel time has been distributed through batches in run,
ing to 2.3 batch

drawing bottoms on 1242, do

1248

11700
11400

2.2
0.25

0.15
6.7

10~

Presert Avergos Operation
1254 1260 1262 1242
12400 13100 13400
12100 12400 12700
2.6 3.0 3.0 1.5
0.25 8.25 0.25 | ©.15
6.2 6.3 6.4 3.78
0.2 0.3 0.3 0.15
9.25 9.65 9.95 | 5.55
9.6 9.3 9.3 |14

on 1260 - 1262,

(11) Semi-Continuous Operetion Jon 1242 only)
Prasent

Initiel cherge 16 11000
Recovery par cycle 1b 65000
Time cycle (tises in hours)

Charge/hest up 2.0
Recycle 0.25
Distilletion * 36.0
8ottoms withdresal 0.5
Cycle time 38.75

Capacity ® 87% 0.8.T. ® 1b/an, 12,3
S0X 1242 on semi-continucus R lh/ 10

18.4
(1967 Product Retio) 14

9.35 (1967 Product Ratio)

1248

1.7
0:15
4.0
8.15

13.9

Jarget (Intermediate)
1254 1260 1262

same

SONe
2.0 2.5 2.5
0.2 0.2 0.2
5.0 6.0 6.1
0.2 0.3 0.3
T.4 9.0 9.1

12.0 10,2 10.2

13.3 (1968 Product Ratio)

11000
65000

1.5

0.15
30.0

0.8
31.18

* Assumes 6 receivers ennoctoa befors boil —down and bottoms withdrewel.

268960 SNOW

Normelly approximately 6 batches are distilled befors
Capscity increase obtained by operation st meximum
temperature differsnces with officient entreineent separetion to prevent product colour degradation.

(1968 Product Retio)




TREATIENT

Cherge we per Distillation Batch.

1282 1248 1234 L8018
Charge time hrs 0.2 0.2 0.2 0.2 0.2
Cool ‘esrth addition 0.3 0.3 0.3 a.3 0.3
Trestment time 2,7 2.7 2,7 2.7 2.7
Filtration 0.4 0.4 0.4 0.4 0.4
Transfer te head tank 1.0 1.0 1.4 1.5 1.8
Cycle time 4.6 4.6 5.0 5.2 8.2
Cepacity ® 98% 0.5.T M 1b 19.2 20.4 20.0 19.8 20.0

Nith the present method of cperstion (single sarth trestment)
cepacity is sdequate for the forscest period. The whole subject of
treatment is under review and Porocel trisle may result in recommendations
for s process ssendment. In ths light of this development,modificetions
to the eyetem will be considered.

b) Pyroclor B
4267 1368 1362 1970 ML 1912

Sales Ferscast W 1b 1.48 1,90 2,38 2.90 3,40 3,68
Stenderd cepecitys 2.4 W 1b.
Pyroclaz blending aperstions will.bs transferred to one of the

internal Arcclor storsge tenks which will be equipped with egitstion.
The incresss of befsh esze will give cepecity pest 1972,

Completisnt 1/89
Cepital o 1,000
c) Aishewsl 1367 M908 A96R 1970 T 312

Mequire- 7.8 8.6 9.4 10.5 11.5 12.5
Pl b 9

™he cepacity of the lesd pot unite is 5.5 W 1b. Installetion
of the prepwsed tubuler unit ie under review. Operation to produce 14 M lb
annum o8 mintmaw Benzene usege (appronimstsly 10X conversion) ie envisaged.
tncresesd capecity s sveileble st ths expense of yleld. Capital cost for the
pro jget ie £305,000. Savings evsilshle (over operstion of lesd pots at
5.5 M 1b level with purchsse of excess from M.(.) acer- .

1989 1970 7L A972
£96,000 £1273000 £1<6000 £171000
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Produotion of Aroclor of sufficient quelity for the Europsen export
market {in particular, Scendinevisn & Italy). continues to be s metter of
concern. Although improvement of power fector ie the most important goel, in
generel the properties of M,C.lL. material aers inferior to competition. The
principle competition is from Beysr, Germeny ( Clophen A 30 and A S0 squivelent
to 1242 end 1254 respectively) and Prodelec, Frence (Pyralene 1499 and 1476)
of the two Beyer producse the supsriar meterial with power factor 0.002

Quelity goals sst up et the $t. Louie Quelity Meeting (Feb. 1968)
These ars considersbly in excess of current produetion capebility.

2roperty —Ggal Newoort (fvoload)  Competition
Power Factor 0.002 0,018 0.002
100°C 30 c/e
5000
Volume Restivity 10,000 7,000 -15,000
]l
Water Content ppm 20 30 20
Colour 10 30 <%
Stability (TCC) Gl - 0.07
0.2

Three srses in which oﬂ'ott 18 being made to improve elsctricel quality
are involved.

(1) Menufecturs of Arccler of low PF in etorage.

Mejor Newpart effort is concentreted on improving nrth trutmt
techniques. Following M.C. findings on the effactive of
(sctiveted Beuxite) lab. tests wers conducted et Newport. HResults eonnmd th-t
Porocel is superior to Fullers Lerth.

Snirsatad =~ Pemgcel  Lazth
Argaler 1242 Ireatment Ioeatesnt

Power Fector
50 C/s 100°C 0,087 0.006 0.045

Tests on snother type of Alumine (Spence) showed no significent
{mprovement over sarth trastment.
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An extendsd plant trial ie scheduled to be run in Februery 1967 using
30/80 mesh Porocsl in the present trestesnt procedure.

In line with the policy eof handling electricel Arsclore in aluminium,
inereesed Arecler 1242 storsge rsquirements are being inetalled in this meterial.
The squipment ceneiste of s tein compertment 12,000 gellon tank ssrved by e new
alusiniue piping system. Replacemant ef the beich treetment tenk,heed tenk end
process piping in eluminium (st o rapitel coet ef 16,000) will be progressad if
found necsseery to mest quality goels.

(2) Prevention of degradetion during loeding and shipping.

Phesed with the installstion of the increesed storege are improved
losding fecilities to hendle the Yo Bine end tenker treffic. A sheeted
building is schaduled fer erection in Februery 1967 shioh sill eliminete
oconteminetion introducsd during loeding in poor weether conditiens.

(3) 1Imp t of sment precielon.

A R,T.8, designed oell suitsble for teeting meterisl in drume or
Tote Bins has besn shewn te functien setisfectorily in lab. trisle en e drum of
Arscler 1242, A CeAeRe 19 in praperation for the previaion of s new meesuring
bridgee Tmeting of eut-geing meterisl in eitu ie expected te yield groster
precieien dus te eliminatien sf centaminetion during esmpling. A relisble
tempereture cerveletion will be required ta meke full use of ‘he installetion.

A.C. resulte have indlceted thet stebility improvements ere also geined
from Porevel trestment. Alternetive methods proposed include KON weehing
of the crude and frectienetion. No work et Newport in thie direotion ie proposed
ot present.

(o)  Eypaglex
Thers ere no major quality probleme essocieted with the Pyrocler output.
(o) pishendd

;Hn quelity prebleme ere invelved.

s.& & Sallutien and 3afety

Thetw ars ne mejor pellutisn preblems in opsratiens sesocisted with
these preducte.

The materiale invelved in Diphenyl, Arecler end Pyrocler Production
de present sefety censiderstione, Hydrogen preduced from ‘he conversion unite ie et
present comp d end exheusted te stmosphere. Although flame t ore
instslled, this represents a potential hszerd. Hydregen praduoed in Tubuler Unit
operstions sill be burned in the furnace.

Minor benzens leaks ers sesocisted eith the inadequats cepecity of the
hydregen compressor. BSenzene hae e toxic ratinge This problem will almso be
elimineted by inetallation of the tubuler unit.
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The installetion of fecilitise to burn Sentowax for stesm jeneratlo.:
will reduos the pressnt hendling riske esssciated with open pena of the molis-
materisl,

Prolanged exposure to Aroclors csh leed to skin disorders. To
eliminets risks from spillagee protective clothing is provided.

The P.P.0. ges scevenger used in Pyroclor production is e skin isritent.
Adoition to the blend terk is et present s meanual operation. Fecilities are
being instelled for transferring the meterisl from Certors to the blend vessel
which will eliminste the riske of contsct to the operstor.
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() Azagiage’
(1) Canvegeion Cast,

Stendard Unit Coet Total Coet for % of Total

(4/000 1b) 1967 et Standerd Conversion
Eate. Coot.
Steam 297 £9,932 15.3
Electricity 32 1,237 1.9
Process Watsr 44 1,700 2.8
Cooling Water 18 696 1.1
Fuel 041 <26 3,710 5,7
447 17,273
Lebour 216 8,348 12.9
Lebour Overhesds 43 1,662 2.8
Supplies 10 a7 0.8
Containers S 194 0.3
Supervision 61 2,387 3.6
Rechenical Cxpense 336 12,988 20.0
Analyticsl Lsb 39 1,807 2.3
7.3.0. . 25 1.5
TOTAL DIRECT 1,102 45,679
1OLRECT 338 12,946 20.0
OEPRECIATION 163 62299 9.7
TOTAL INDIRECTS 498 19,245
TOTAL CONVERSION COSTY. 1,682 64,926
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(1) -ASDUCTION COST BY GRADE

1242 l‘” ;ig 2 62
(o) () (e)| (o) b fc)}] (o) (b (c)] (a) (b {c)] (o) ﬁl_ c)
12 £10 0.4) 12 2,5 0.14

Carth 12 £310 0.13 12 £31.3 0.13 12 £110 0.13

Lise 3 77.5 0.03 3 7.8 0.03 3 27.5 0.03 3 2.5 0.03 3 0.63 0.03

Felly L] 129 0.06 L] 13.0 0.06 S 45.8 0.06 5 4.17 0.06 5 1.04 0.06

Chlorins 2716 38.4 {3109 6096 35.0 | 3500 32083 239.6 | 3905 3252 44.5 | 4010 835 45.8
3037 633 34.8

Oiphenyl 4515 116 '58,S | 4078 10620 45.9 | 3626 33238 4l.1 | 3182 2652 36.2
Conversion 1682 _4345] 18,9 11682 4300 10,9 11602 1%5410 19.]1 | 1662 1402 19.1 | )682 350 19,2

Totel tve. Cost8933 230490 8809 23148 8828 80922 8789 7325 8749 1822

HCl Credit 414R 1069SR 4.6 |474R 1234 5,3 | 833 48868 6.0 | S9SR 496R 6.8 | 610R 1278 7.0

Total Pred, 8519 21979% 8415 21914 6298 76036 8194 6829 6139 1695
Cost —
Column (a) Stenderd Unit Cest d/000 1b

Column (b) Annual cost Por 1967 et the standard rete

Column (c) Percentage of Total Rosttive Cost

ood
]
Q@
d
"
Q
«2
r 3
Q
X
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(b)  yrpclor BP0,
Stendard Unit Cost ToZel Cost for '67 % of Totsl
\J R

+ Larth 2 do

0.02
Alumine 34 154 0.3
TC8 4386 19882 40,2
1260 . 508C 23027 46,5
RO 208 4380 2.7
2802 Add32
Stoon 167 757 1.5
Electricity 4 -2 0.04
17 775

Lebaur 1086 843 1.7
Lebeur Overheeds n 172 0.4
Supplies 18 82 0.2
Supsrvisien 23 104 0,2
Mech. Exp, 3 33 .7
Ansl. Leb, 128 567 1.1
Te8.0. A8 218 0.4

TOTAL OIRECT 10,484 47,524

INDIRECT 319 1,448 3.0

DEPRECIATION 1.0

TOTAL INDIRECTS

TOTAL CoST R 42,490

MONS 056839



b) Sgeciel Pyroclos

Stenderd Unit Coest Total Cost for fLof T |}

(a/000 1b) 1967 ot etenderd Produc .n
e fatts —
Carth 46 69 0.4
TC8 2,748 4,144 2%.8
1254 6,308 9,506 9.2
9,098 13,7119
Steem 112 169 1.1
Electricity A8 —3 0.2
m 192
Lebour 285 400 2.%
Lebour Overheads 53 80 0.5
Suppliss 30 45 0.3
Supervision 34 S1 0.3
fechanicel Expenes 110 166 1.0
Analyticel Lab 190 287 1.8
T.8.0 . P/ 3 108 - D2,
ToTAL otaEtt 9,979 15,049
Indirect 408 733 4.8
Deprecistion -1 1.6
TOTAL INOIRECT [L[}} 997
TOTAL COST 10,640 16,048
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Berzene
co.

led
Acatone

Stoom
Clectricity
Process Watar
Cooling water
Fuel 041

Labour

Labour Overhesds
Supplies
Containers
Supsrvision
Mechenicel Expense
Anslyticel Leb
T.3.0

TOTAL OIRECT
Indirect
Osprecistion
TOTAL INDIRECT

TOTAL COST

-19 -

Standerd Unit Coet

Totel Cost for 1967

3
88,047

527

596

90,040

5,363
1,008

413
1,673

pEN H
22,870

8,319
1,673 .
6
183
1,873
13,911
76

2,406
141,787

11,848
.22
25,140

166,927

£ of Total
B

..

—Oo@=0O0~n
. .o
SPRAGGO~VLOO
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1.
2.

3o

4

7.

Isprove Quality

Install increseed
sterags & bulk
leading fecilitise

Instell Brinks &
Cxternel cooling
fecilities on
chlorinstors

Incresse distill-
ation cepscity

Alternate (1)
Alternete (2)

In.line testing
fecilitien

Impreved Pyresler
handl

fecillities
Tubuler Resctsr

& 1,000

£308,000

Quality
Quality/
nest
requirements

Nest
roquirements

Safety

40 6/87
20 4/87
80 1/68
80 4/87
(Interwed).
(1) 1/68
(11) 8/68
20 4/67
10 yer

MONS 056842
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O.  PEGPOAT ON PAZVIOUS GOMLD
1)  guality Isorovessnt
A plent trial of deuble earth trestment yhldod no uw\!.nmt
d ehow 1

increess in quality. An EVOP programme run en single earth ¢ t
product quality snd the eptimum levels ssteblished ers now heing mrohd.

As o result of tziale on drume, two madificetions te the F.P.
packegs have besn intreduced.

(1) A superier lecquer based sn e 30K epoxy 70K phendide: hes
been introduoed which cembines resistance te power f.
degredetion with grester mechanicsl etrength. prevent
deterioretion of the pnd\ne we uwu of flek 1ining

through maltreatment in shipment.

(2) Additien ef serth te the F.P. peckage ea @& shert term msthed of
guarentesing the quality of meterisl en srrivel et the oustemer.
Trisl drume were set te I.CoAsR. Salee Department report that
the custemer has fsund this matsrial satisfectery.

2)  Iubulax Resctes

CoAoRe submission wes dalsyed dus te re-svalustisn by
C.E.D. of the ocapital requiremente.

3)  Inaceass Arsclors aceduchisn casscity $@ 10 @ Ab

Installetion of o 4" ghlerine supply line in Nevesber
has allowsd eperatisn of chlerineters et the requirsd level.

M) Incresse dishenyl capacity te 5:5 ® 2b

1966 productien figuree show that @ ocepecity of 5.33 M lbe
was schieved in 1966 pered ts 5.13 M 1bs, in 1968 ,

S)  JIngxasee giphwoyl cacacitv te 6.3 R ib '
Thie preject was fer installetion of mo0journ pote on the
tws sxisting units. This preject wee not cerried out swing to likelihoced ef
spprevel being given ts the Tubuler Unit Preject.

§)  lvdceosn Recovery

Investigetion showed hydrogen recovery te be nen-feseible.
Ne methode exiet fer an ecenomic cleen-up te levele at which use oould be mede
of the hydrogen by ether plante.

7)  Plont Resistivity Meter .
A cell suitebls fer in-line resistivity end pewer fecter
messuresmente was designad by M.T.8. Succeseful in-drum teste were carried eut
in December.

CJSP/ER, 3.2.67

é‘,jqﬂ,,y.
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